Author's response to reviews

Title: Assessment of data quality in a multi-centre cross-sectional study of participation and quality of life of children with cerebral palsy

Authors:

Heather Dickinson (heather.dickinson@ncl.ac.uk)
Kathryn Parkinson (kathryn.parkinson@ncl.ac.uk)
Vicki McManus (vmcmanus@enableireland.ie)
Catherine Arnaud (carnaud@cict.fr)
Eva Beckung (eva.beckung@vgregion.se)
Jerome Fauconnier (jerome.fauconnier@imag.fr)
Susan I Michelsen (sim@si-folkesundhed.dk)
Jackie Parkes (j.parkes@qub.ac.uk)
Giorgio Schirripa (deceased@deceased.it)
Ute Thyen (thyen@paedia.ukl.mu-luebeck.de)
Allan Colver (allan.colver@ncl.ac.uk)

Version: 4  Date: 31 October 2006

Author's response to reviews:

We would like to keep Table 3 as it is as all the data on "relative risk functions" are in it. However we agree we should not call it Table 3(a) etc. We have therefore always referred to it as Table 3 but then followed this by section (a) or section (b) or section (c). I hope you are agreeable to this.

As I should have done in the previous version, I have removed the references in the text to additional files containing Tables.